Author's response to reviews

Title: Multidisciplinary Approaches Vary as to the Role of Endoscopic Ultrasound (EUS) - Guided Fine Needle Aspiration: Controversial Indications

Authors:

Jesse Lachter (J_Lachter@rambam.health.Gov.IL)
Yoav Rosenthal (Yoav.Rosenthal@gmail.com)
Yoram Kluger (Y_Kluger@rambam.health.Gov.IL)

Version: 3 Date: 27 June 2011

Author's response to reviews: see over
Dear Editors,

Thank you very much for the peer-reviewing and comments which you sent regarding our manuscript. Having read your comments carefully, and re-reading the manuscript, we have found validity to all of your many ideas. The manuscript has been fully re-worked to include all of the changes suggested, from the title to the authoring, the formatting, and the improved review of consensus sources and discussion of risks, and of course the English has been proofread and polished.

We hope that it now meets the high standards and after these many improvements is acceptable for publication.

The manuscript is being forwarded to you with track changes so that you may see the changes which have been made, as per your instructions.

Thanking you once again for all editorial and peer-reviewing ideas and improvements.

Sincerely,

Jesse Lachter
Yoav Rosenthal
Yoram Kluger

Encl are two manuscript body files: one is with the "track changes", the other is after changes were accepted.

Title
A Multidisciplinary Survey on Controversies in the Use of EUS-guided FNA: assessing perspectives of surgeons, oncologists and gastroenterologists

Authors:
All are from
Rambam Health Care Campus
Bat Galim, Haifa, Israel
And
Technion- Israel Institute of Technology, Faculty of Medicine

Jesse Lachter MD (corresponding author) - Department of Gastroenterology
J_Lachter@rambam.health.Gov.IL

Yoav Rosenthal MD- Department of Gastroenterology
Yoav.Rosenthal@gmail.com

Yoram Kluger- MD- Department of Surgery
Y_Kluger@rambam.health.Gov.IL

Contact author address: Hantke 36
Haifa, Israel 34608
e-mail J_Lachter@rambam.health.Gov.IL
Tel 01197248247826

The authors hereby assert that there are no conflicts of interest and no disclosures to make, and that this study was not funded in any way.